European Journal of Educational Sciences

## **REVIEW HISTORY**

Paper: "Free Education Policy and its Emerging Challenges in Sri Lanka"

Corresponding Author: A. H. G. K Karunaratne Email: <u>kingsley@sjp.ac.lk</u>

Doi: 10.19044/ejes.v7no1a1

Peer review: Reviewer 1: Blinded

Reviewer 2: Blinded

Published: 31.03.2020

